Saving Teens @

Using a Policy Discontinuity to Estimate the Effects of Medicaid Eligibility

Laura R. Wherry Bruce D. Meyer

ABSTRACT

We examine the immediate and longer-term mortality effects of childhood public health insurance eligibility during childhood. Our identification exploits expansions in Medicaid eligibility that applied only to children born after September 30, 1983. This feature resulted in a large discontinuity in the cumulative years of eligibility of children at this birth date cutoff. Under the expansions, black children gained twice the years of Medicaid eligibility as white children. We find a later-life decline in the rate of disease-related mortality for black cohorts born after the cutoff. We find no evidence of a similar mortality improvement for white children.

THE JOURNAL OF HUMAN RESOURCES • 51 • 3

Laura R. Wherry is an assistant professor in residence at the David Geffen School of Medicine at the University of California, Los Angeles. Bruce D. Meyer is a McCormick Foundation Professor of Public Policy at the Harris School of Public Policy at the University of Chicago and research associate at the National Bureau of Economic Research. The authors would like to thank Gary Freed, Andrew Goodman-Bacon, Robert Kaestner, Miriam Kalichman, Genevieve Kenney, Lindsey Leininger, Tony LoSasso, Sarah Miller, Karin Rhodes, Dan Schmierer, and participants at seminars at the Georgia Institute of Technology, the 2011 Census Research Data Center Conference, Cornell University, Purdue University, Syracuse University, University of Michigan, the 2011 Midwest Health Economics Conference, and many additional seminars, for helpful comments. They are also grateful to Negasi Beyene, Clint Carter, Karen Davis, Adrienne Jones, Frank Limehouse, Brian Littenberg, and John Sullivan for assistance in accessing the restricted data used in this project at the Chicago Federal Reserve, University of Michigan, University of California, Los Angeles, and University of Southern California Census Research Data Centers. Wherry acknowledges research support from the Agency for Health Care Research and Quality National Training Fellowship in Health Services Research, the Robert Wood Johnson Foundation Health & Society Scholars program, and the California Center for Population Research. Mever acknowledges research support from Robert Wood Johnson Foundation Grant Number 043647, the Searle Fund, and the NICHD through R01 HD049580. The computer programs and publicly available data used in this article can be obtained beginning six months after publication through three years hence from Laura R. Wherry, David Geffen School of Medicine at University of California, Los Angeles, LWherry@mednet.ucla.edu. The restricted use data, specifically the 1979-2012 Multiple Cause of Death files

LWherry@mednet.ucla.edu. The restricted use data, specifically the 1979–2012 Multiple Cause of Death files with month and year of decedent birth (both restricted) as well as state and county of residence (also restricted starting in 2005), can be obtained from the National Center for Health Statistics. [Submitted September 2013; accepted April 2015]

ISSN 0022-166X E-ISSN 1548-8004 © 2016 by the Board of Regents of the University of Wisconsin System Supplementary materials are freely available online at: http://uwpress.wisc.edu/journals/journals/ jhr-supplementary.html

I. Introduction

Interest in public health insurance in the United States is in part motivated by the relatively poor health of children in families with low incomes. Not only are children from poor families more likely to be born in poor health, but the relative health of these children worsens with age (Case, Lubotsky, and Paxson 2002). To address this disparity in child health, U.S. policy has largely focused on increasing access to medical care for children through expansions in publicly provided health insurance.¹ These expansions aim to promote health among children by providing timely access to health services. A major focus is preventive care and the early diagnosis and treatment of physical and mental health needs with aims to limit pervasive health problems in adulthood (Rosenbaum et al. 2005).

There is strong evidence that public health insurance coverage for children increases their utilization of medical care, but the evidence that this leads to improved health outcomes is much more limited.² The strongest evidence of health improvements is seen in contemporaneous declines in child mortality associated with expanded public health insurance eligibility. In two seminal papers, Currie and Gruber (1996a, 1996b) find that Medicaid expansions in the mid-1980s and early 1990s reduced mortality among both infants and children aged one to 14, with declines in child deaths where the underlying cause was disease related (internal). The authors find no effect on child deaths due to other (external) causes, such as accidents or homicides. Howell et al. (2010) examine expansions in public insurance for children aged one to 17 occurring over a longer time period and find additional evidence of a decline in child mortality although concentrated among deaths due to external causes.

Until recently, there has been little attention to the long-term effects of public health insurance coverage for children. Yet many pediatric health services are designed to protect healthy children from future disease or risks, such as immunizations and health education, and the payoffs from these types of services might not be evident until later in life. Furthermore, for those children with mental and physical disabilities or chronic illness, early diagnosis and treatment may help the management of these conditions or limit their persistence into adulthood.³ From a program evaluation standpoint, the effect of public coverage on the long-term health of children is an important component of program benefits.

^{1.} A series of expansions of the Medicaid program beginning in 1984, and later expansions under the State Children's Health Insurance Program (SCHIP), are estimated to have more than doubled the share of children eligible for public health insurance from 17.8 percent in 1987 (Cutler and Gruber 1996) to 41.7 percent in 2000 (LoSasso and Buchmueller 2004).

^{2.} Several papers have documented an increase in annual doctor visits (Currie and Gruber 1996a, Card and Shore-Sheppard 2004, Currie et al. 2008) and hospital use (Currie and Gruber 1996a, Dafny and Gruber 2005) under expansions in public health insurance for children. Yet, a number of studies using parental reports of child health measures, including child health status and activity limitations, find no evidence of improvement under public health insurance (for example, Currie and Gruber 1995; Racine et al. 2001; Currie, Decker, and Lin 2008). Howell and Kenney (2012) provide a comprehensive review.

^{3.} Medical care may also have a positive impact through counseling regarding behavior risks and linking children to helpful social services. The standard economic argument is also applicable that providing free medical care may free up resources for making other investments with long-term impacts.

A small but growing literature considers the longer-term health effects of public health insurance for children. Two published papers examine later-life health using self-reported measures of health and with mixed findings. Using within-state variation in past eligibility for different birth cohorts, Currie, Decker, and Lin (2008) find evidence of better self-reported health status at ages nine to 17 among cohorts in states with greater eligibility for public health insurance at ages two, three, and four. De la Mata (2012) examines the longer-term effects of public eligibility by comparing health outcomes among children aged five to 18 in families with incomes just above and below the income eligibility thresholds for public health insurance. Examining outcomes five years after a given year of eligibility, she finds no evidence of significant effects on child health status, obesity, and the number of school days missed due to illness. More recently, a number of new working papers find evidence linking Medicaid eligibility during childhood with improved health (Boudreaux, Golberstein, and McAlpine 2014; Miller and Wherry 2014) or economic outcomes at later ages, including higher rates of high school and college completion (Cohodes et al. 2014) and improved labor market outcomes (Brown, Kowalski, and Lurie 2015). Each of these papers relies on variation within states over time in the generosity of Medicaid eligibility rules and assumes exogeneity in the timing of state eligibility changes.

This paper provides new evidence on the immediate and longer-term health effects of public health insurance during childhood using a more clearly exogenous source of variation than the existing literature. Our identification strategy exploits a unique feature of several early Medicaid expansions that extended eligibility only to children born after September 30, 1983. The change in eligibility occurred when these cohorts were almost eight years old, making it impossible that families could have altered their delivery dates in anticipation of the change. This policy discontinuity was first identified and used by Card and Shore-Sheppard (2004) to examine changes in insurance coverage and medical care utilization under the expansions.⁴

Building on their pathbreaking work, we show that this eligibility rule led to a large *cumulative* difference in public health insurance eligibility for children born before and after September 30, 1983. Children in families with incomes at or just below the poverty line gained close to five additional years of eligibility if they were born in October 1983 rather than just one month before. This gain in eligibility occurred between the ages of eight and 14. Using regression discontinuity methods, we examine mortality at these ages for cohorts of children born just before and after the birth date cutoff to identify the immediate impact of this gain in public health insurance eligibility.

Importantly, our research design also allows us to examine the impact of childhood Medicaid eligibility later in life by following cohorts born before and after the cutoff through age 23 to observe any *longer-term* impacts on mortality. We examine changes in mortality at the ages of 15–18 and 19–23 when cohorts on either side of the cutoff were equally likely to be eligible for public health insurance. This allows us

^{4.} The authors found an increase in coverage among newly eligible children born after this date with little evidence of crowdout of private insurance. They also provide evidence that this new Medicaid coverage led to a large increase in the utilization of medical care.

to attribute any changes in mortality we observe at these ages to differences in Medicaid eligibility that occurred earlier in childhood.

As Lee and Lemieux (2010) have emphasized, the regression discontinuity methods that we employ can be just "as good as a randomized experiment." Although not an experiment, our research design approximates random assignment in public health insurance for children, a group for which there has been no randomized trials of private health insurance coverage or Medicaid.⁵ And with this variation occurring nationwide, it offers advantages in terms of the generalizability of findings, larger sample sizes, and the ability to examine long-term impacts.⁶ For these reasons, the results of this policy experiment merit special attention.

Following the literature, we examine rates of mortality by the underlying cause of death, distinguishing between deaths due to internal and external causes in our analysis. As mentioned earlier, both types of death have been linked to changes in public insurance coverage. We also examine changes in mortality separately by child race since black children were particularly likely to benefit from the Medicaid expansions due to their distribution of family income. Existing evidence also suggests that expansions in public insurance were more likely to affect black children.⁷

We find compelling evidence of a sizeable decline in the later-life mortality of black children under the Medicaid expansions. The regression estimates indicate a significant decrease in the internal mortality rate for black children born after September 30, 1983, at ages 15–18. The improvement in mortality for black children is not reversed during the early adult years at ages 19–23, although there are no clear additional gains at these ages resulting from childhood eligibility. We find no evidence of a similar decline in mortality for white children under the expansions.

This paper is structured as follows: Section II provides some background on the Medicaid expansions that resulted in discontinuous childhood eligibility at the September 30, 1983, birth date cutoff. Section III reviews our research design and data. Section IV presents our results, while Section V interprets them. Section VI concludes with a summary of our findings and their applicability to public policy.

^{5.} The RAND Health Insurance Experiment randomized people into plans with different levels of cost sharing but provided all participants with health insurance. Researchers found no significant differences in health measures for children in families randomly assigned to health insurance plans providing medical care free of charge and those requiring some form of copayment despite reduced demand for ambulatory care for children in families enrolled in the cost-sharing plans (Valdez et al. 1985). Some gains in health were observed among poor adults with existing chronic health problems, while individuals between the ages 12 and 35 on the free-care plan showed some improvement in dental health (Manning et al. 1987).

^{6.} For instance, while the Oregon Health Insurance Experiment found improvements in self-reported health and depression among adults gaining Medicaid coverage, it had less statistical power to detect changes in outcomes with lower prevalence including mortality and specific physical measures of health (Finkelstein et al. 2012, Baicker et al. 2013).

^{7.} Currie and Gruber (1995) find a much larger mortality decline among black children than white children (more than 4× the size) in their analysis, although the changes are not statistically significant for either group. A recent paper by Goodman-Bacon (2014) finds large declines in nonwhite infant and child mortality following the introduction of Medicaid in the 1960s and 1970s. He finds no evidence of similar improvements in mortality for white infants and children. Howell et al. (2010) report no significant difference between black and white children in the effect of the eligibility expansions, which they test by including a black race interaction term in their model. The point estimate is not reported, however, making it difficult to gauge whether the estimate is small and precisely estimated or potentially large and imprecise.

II. The Discontinuity in Eligibility

Medicaid is a joint federal-state program that provides health services to certain low-income beneficiaries. Since 1967, the program has included a special benefit package for children designed to provide comprehensive health coverage including preventive care. This benefit package provides all Medicaid children with early and periodic screenings for physical, mental, and developmental health; vision, dental, and hearing examinations and followup care; and all health services they are found to need. Importantly, children with eligibility for Medicaid have "conditional coverage" in that, even if they do not actively enroll in the program, their expenses are covered in the event of hospitalization or high health care spending (Cutler and Gruber 1996).

Historically, not all children in poor families were eligible for Medicaid coverage. Eligibility for nondisabled children was previously linked to participation in the Aid to Families with Dependent Children (AFDC) program. Beginning in 1984, a series of Congressional acts expanded eligibility for Medicaid to children who were not traditionally eligible for AFDC, often with income levels above AFDC cutoffs. To phase in the Medicaid expansions, many of the legislative changes were applied only to children born after September 30, 1983 (see Table A1 in the online appendices for additional details).⁸ This legislative provision meant that children born in October 1983 faced more generous eligibility criteria for the Medicaid program than children born just one month earlier in September 1983. The result of this provision was a large discontinuity in the number of childhood years of Medicaid eligibility for children born around this birth date cutoff.

To illustrate the discontinuity, we model childhood public eligibility for monthly cohorts born within four years of the birth date cutoff—between October 1979 and September 1987. Using detailed federal and state eligibility rules for the years 1979–2005, our simulation program uses information on state of residence, family structure and size, parent employment, and family income to calculate monthly public eligibility status through age 17.⁹ To obtain a nationally representative sample of these characteristics of children and families, we employ data from the March Supplements to the Current Population Survey (CPS). We use a random sample of 500 children of ages zero to 17 from each year of the 1981–88 CPS and estimate the childhood eligibility for this pooled sample for each birth month.¹⁰

Figure 1 shows the results of our simulation for children in families with incomes below 150 percent of poverty, those most likely to be affected by the change in Medicaid eligibility rules. This graph depicts the average total number of years of public eligibility during childhood by birth month cohort. Each line on the graph represents an income

^{8.} The appendices referred to in the text may be found online at http://jhr.uwpress.org/.

^{9.} For the years prior to welfare reform, we estimate eligibility for Medicaid under AFDC, state Ribicoff rules, and federal and state Medicaid expansions. For 1997 forward, eligibility is estimated under the postwelfare reform eligibility rules for Medicaid, as well as under continuing state Medicaid expansions and new separate state programs funded by SCHIP. See Appendix A for additional details.

^{10.} We use the 1981–88 CPS surveys because the information collected on income is for the previous calendar year. This simulation holds family characteristics constant over the child's lifetime. This convention may understate eligibility during early childhood and overstate eligibility during the later years of childhood if family income grows with child age and is not counterbalanced by changes in family size that alter the poverty thresholds to which income is compared.



information. Family income is indexed using the CPI-U and assumed to be constant over the child's lifetime.

Table 1

Childhood Medicaid Eligibility Gain for Children Born in October versus September 1983 by Child Race

	Average Gain	Percent	Average Gain (in Years)
	(in Years)	Gaining	for Children Gaining
	for Population	Eligibility	Eligibility
All children	0.42	9.24	4.53
Black children	0.82	17.13	4.81
White children	0.37	8.18	4.48

Note: Weighted average calculated using a sample of children aged zero to 17 in the 1981–88 March CPS and eligibility simulation methods detailed in the text.

bracket. For all income brackets, we see a gradual increase in childhood eligibility across birth cohorts prior to a jump in years of eligibility for children born in October 1983. Children born in October 1983 are the first cohort of children to be eligible for the Medicaid expansions with the September 30, 1983, birth date cutoff. Following this jump, we again see a gradual increase in childhood eligibility for the remaining birth cohorts.¹¹

As depicted on Figure 1, the magnitude of the discontinuity in childhood eligibility at the September 30, 1983, cutoff varies by family income. The largest jump in years of eligibility occurs among children with family incomes above AFDC eligibility levels but below poverty. Children in families with incomes just below the poverty line see the largest gain with an additional 4.6 years of eligibility during childhood, while those with incomes between 50–74 percent of the poverty line see a gain of 3.4 years. We see a smaller discontinuity in childhood eligibility among children in families with incomes at or below AFDC levels, around 50 percent of the poverty line. Many of these children were already eligible for Medicaid through AFDC prior to the expansions.

We examine the effects of the expansions separately for black and white children. At the time of the Medicaid expansions, black children were nearly three times as likely to be in poverty as white children. Therefore, we may see a larger impact of the expansions among black children. Table 1 reports the estimated discontinuity in childhood eligibility at the September 30, 1983, cutoff by child race.¹² We estimate that 17.3 percent of black children gained on average just under five years of eligibility compared to similar gains in eligibility for 8.2 percent of white children. As a result, black children gain, on

Much of this gradual increase is due to a larger share of each successive cohort's life occurring when younger children were Medicaid eligible under various age cutoffs rather than a gradual change in rules.
 To compare the eligibility gains by child race, we use a sample that draws a maximum of 500 children aged zero to 17 from each race group and state cell from the pooled 1981–88 CPS years, giving us 50,874 child observations. Holding family characteristics constant, we estimate eligibility for each child for each year during

childhood by calculating eligibility in July of the appropriate calendar year. We use these estimates to calculate the average gain in years of eligibility at the September 30, 1983, cutoff by child race. In calculating all averages that we report in the text and in tables, we multiply the CPS-provided survey weights by an adjustment factor specific to each cell in order to account for the manner in which we constructed our sample. This factor is calculated as the inverse of the ratio of the sum of the survey weights of individuals in the cell in our sample and the sum of the survey weights of individuals in the cell in the CPS.

average, 0.82 years of eligibility, which is more than twice the average years gained by white children (0.37 years).

To illustrate the timing of the gain in eligibility, Figure 2 compares the share of children eligible for public health insurance if born in October versus September 1983 at each year of age by race. The main legislative source behind the jump in eligibility is the implementation of the Omnibus Reconciliation Act of 1990 (OBRA90). Effective July 1, 1991, OBRA90 mandated state coverage to children younger than age 19 born after September 30, 1983, with family incomes below the poverty line. Children born in October 1983 were almost eight years-old when this change was implemented. For both black and white children, we see a sizeable jump in the share of October-born children with eligibility at age eight. However (as indicated in Table 1), a much larger share of black children is affected by the change in eligibility rules.

Faced with new and more generous eligibility criteria, the October 1983 cohort remains more likely to be eligible for public health insurance until the introduction of the State Children's Health Insurance Program (CHIP) in August 1997. Authorized by the Balanced Budget Act, CHIP allowed states to expand eligibility for public health insurance to children of higher incomes, regardless of the child's month of birth, under existing state Medicaid programs or new state programs. These expansions were quickly adopted by states and served to close any discontinuity in eligibility by age 15.

We distinguish between four different age ranges in our analysis. Ages four to seven capture a period prior to the implementation of OBRA90 where differences in eligibility at the birth date threshold are small. Our analysis of this period serves to investigate any differences in health for children at the cutoff, although this period also captures minor differences in eligibility. Later, we test for baseline differences in the health of these cohorts using measures of infant health and find no evidence of preexisting differences.

Next, we examine children at ages eight to 14. Children who benefitted from the expansions primarily gained eligibility during this age range, as shown in Figure 2. We interpret any discontinuity in health observed during this period as an immediate effect of public health insurance coverage, similar to the existing estimates using cross-sectional data in the literature (for example, Currie and Gruber 1996a). However, any difference in health observed in our estimates may also be due to a cumulative gain in exposure to health insurance coverage over the period. For instance, a child aged 14 born in October 1983 may have gained Medicaid coverage for the five years prior, beginning at age eight. There is little research that has examined the effect of multiyear coverage on health care access or outcomes for children¹³ despite the potential complementarities in investments in health at different ages (Becker 2007).

Finally, we examine children at ages 15–18 and ages 19–23, which are periods following the gain in public eligibility. At these ages, cohorts born on either side of the birth date cutoff were equally likely to be eligible for public health insurance. By examining these older age groups, we are able to consider the longer-term effects of childhood coverage during late adolescence and early adulthood.

^{13.} One exception, Cassedy, Fairbrother, and Newacheck (2008), shows that children with any gap in insurance coverage over a two-year period have higher odds of lacking a usual source of care and not having wellcare visits than those with continuous insurance coverage, regardless of whether that continuous coverage was public or private. There is also a substantial literature finding that continuous coverage during a year is associated with better access to care (see Leininger 2009 for evidence and cites to the literature).



more information. Family income is indexed using the CPI-U and assumed to be constant over the child's lifetime.

564 The Journal of Human Resources

III. The Research Design

Our research design uses comparisons across the September 30, 1983, birth date cutoff to measure the impact of Medicaid eligibility on mortality. Because factors other than date of birth determine childhood eligibility for public health insurance, such as childhood family income, this is an example of a "fuzzy" RD design. Birth cohort is defined in our analysis using the month and year of birth since this was the finest information on date of birth that we were able to acquire in the mortality data. Our sample includes all cohorts born between October 1979 and September 1987. We denote birth cohorts using the integer values $c \in [-48, 47]$, where c=0 for the birth cohort October 1983, the first cohort born after the cutoff date September 30, 1983.

For each outcome variable, we first present graphical evidence to visually check whether there is evidence of a discontinuity at the September 30, 1983, birth date cutoff. We then present regression estimates to confirm the statistical significance of any observed discontinuity and its robustness to the inclusion of covariates. Our main analysis is based on a regression discontinuity (RD) model of the form

(1) $y_{ca} = \alpha + f(\gamma; c) + \beta 1(c \ge 0) + \delta_m M_c + \varepsilon_{ca}$

where y_{ca} represents the mortality rate for cohort c at age $a, f(\cdot)$ is a smooth function of birth month cohort with parameter vector γ , and $1(c \ge 0)$ is an indicator for whether the cohort was born after September 30, 1983. We also include calendar month dummies M_c given the evidence on seasonal variation in maternal characteristics for births and its relation to later outcomes (Buckles and Hungerman 2013).¹⁴ We interpret β as the effect of the eligibility expansions averaged across the full sample of children at the cutoff.¹⁵

The central question for implementation is how to model the function of birth month cohort $f(\cdot)$. We present estimates of the treatment effect using both linear and quadratic specifications of the regression function. We also report estimates where we allow the elements of the parameter vector of these specifications to differ on the two sides of the birth date cutoff (specifically linear and quadratic splines). In addition, we examine the sensitivity of our estimates to the use of narrower windows of observations around the birth date cutoff and present results for four-year ($c \in [-48, 47]$), three-year ($c \in [-36, 35]$), and two-year ($c \in [-24, 23]$) windows of birth month cohorts on either side of the cutoff. All regressions are estimated using OLS.

As described above, we estimate Equation 1 separately for each of the following age groups: four to seven, eight to 14, 15-18, and 19-23. In accompanying graphs, we plot the means of the outcome variable by birth cohort in the four-year observation window separately for each of the four age groups. The lines are fitted values from a regression that includes a second-order polynomial in birth cohort and a dummy variable for children born after September 30, 1983. In our regression results, heteroskedasticityrobust standard errors are clustered by birth month cohort to account for our grouping of narrow age ranges into broader ones, as we describe below.¹⁶

^{14.} The inclusion of calendar month dummies will also account for the influence of other policies that may have differential effects based on month of birth. One example might be birth date cutoffs for school entry. 15. In other words, we estimate the proportion eligible times an intent-to-treat parameter.

^{16.} Clustering on each value of the discrete assignment variable can also account for uncertainty in the choice of the functional form, as shown by Lee and Card (2008).

A. Outcomes and Data

To measure the effects of childhood Medicaid eligibility on immediate and long-term health, we use mortality records for all deaths in the United States. In contrast to parental reported measures, mortality is a clear and objective measure of health. However, examination of this measure is not without its limitations. Given that mortality is a relatively infrequent event among children, any changes in population-level mortality rates are likely to be small and may be difficult to assess. In addition, a focus on mortality may miss other important but difficult to measure effects of public health insurance on the health or quality of life of children. Acknowledging these limitations, reduced mortality remains an unambiguous and important outcome of expanded health insurance coverage.

Mortality is also an outcome for which potentially we might also see longer-term impacts of health insurance coverage for children. Children face lower mortality rates during the middle childhood years and higher rates of death in early and late childhood.¹⁷ Given this relationship, the impact of childhood coverage on underlying health might be more apparent during the adolescent years when mortality is higher.

We construct aggregate mortality using information from the National Vital Statistics System (NVSS) Multiple Cause of Death files for the years 1979–2011. We construct cohort-specific death rates, where cohort is defined as the month and year of birth. While the date of birth of the decedent is excluded from the publicly available mortality files, we are able to access birth month and year as restricted information at Census Research Data Centers. Necessarily, we drop all observations from the mortality data that is missing information on child birth month or year. The average share of child observations with missing birth date information for each year of mortality data is about 2 percent. We also exclude those observations where the birth year information contradicts the reported age of the child. In addition, we use only observations for children born in the United States with residence in the 50 U.S. states and the District of Columbia. None of these restrictions are discontinuous at the September 30, 1983, cutoff and our results do not change when using an unrestricted sample (see sensitivity analyses in online Appendix B).

For our outcome measure, we construct the rate of death for each of the 96 birthmonth cohorts in our sample for the following age subgroups: four to five, six to seven, eight to ten, 11-12, 13-14, 15-16, 17-18, 19-20, and 21-23. The observations for each of our four age group regressions include the 96 cohorts times the two or three age subgroups that make up the four age ranges described earlier. The definition of these four wider age groups was motivated by the ages affected by the expansions as shown in Figure 2. To construct rates of death, we tally the total number of deaths for each birth cohort *c* at specified years of age *a*. We then divide this number by the total population at risk for each birth cohort and age, defined as the total population of births for each birth cohort minus the total number of deaths for all ages prior to age *a*.¹⁸ We construct

^{17.} Figure A1 shows the age profile of rates of mortality due to internal and external underlying causes for children in their first year of life through age 19 in the year preceding the Medicaid expansions. For both types of death, the relationship with child age takes on a U-shape, though it is less pronounced for internal causes. 18. We obtain birth counts from the NVSS Birth Data files for the years 1979–87. The 1979–84 files offer only a 50 percent random sample of births for a handful of states, stratified by month of birth and county of occurrence. For each of these states and years, we double the number of births for each birth cohort to approximate the universe of births. We make this adjustment for the following states and years: Ariz., Calif., Del., Ga., and D.C. in 1979–84, N.D. in 1979–82, N.M. in 1979–81, and Ark. and S.D. in 1979.

separate outcomes for black and white children and express all final rates as the number of deaths per 10,000 children in each birth cohort. In sensitivity analyses, we examine regressions using the numerator only (number of deaths for each birth cohort) as an alternative dependent variable, with no meaningful impact on our findings.¹⁹

In our results, we distinguish between internal and external causes as the underlying cause of death, which we classify according to the medical code reported on the death certificate. For the years 1999–2011, the cause of death is reported according to the Tenth Revision of the International Classification of Diseases (ICD-10), while the Ninth Revision (ICD-9) is used for the <u>1980–98</u> years. Because the two classifications differ slightly, we include an ICD-10 dummy in our regression specifications to account for this change.²⁰

We also examine more narrow classifications of the underlying causes of death. For internal causes, we code subcategories following the chapter structure of the ICD-10.²¹ We code subcategories of external causes following the groupings in a tabulation list developed by the National Center for Health Statistics for use with mortality data classified by ICD-10.²² To aid in constructing comparable measures over time, we draw on the assignment of ICD-9 codes to the ICD-10 tabulation list presented in Anderson et al. (2001).²³

Table 2 presents the average rates of death for cohorts born prior to September 30, 1983, in our sample by race and age group. Mortality is much higher among older children across all groups, with mortality rates for 19-23-year-olds almost four times as high as those of children ages four to seven. Although deaths due to internal causes are more common among older children, it is the frequency of deaths due to external causes that drives most of the increase in total mortality for the older age groups. Black children have higher rates of mortality than white children across all age groups.

Among all children, the most common causes of internal death are neoplasms (tumors), diseases of the nervous system (which include meningitis, epilepsy, and cerebral palsy), cardiovascular diseases, respiratory diseases, and congenital malformations and chromosomal abnormalities. With the exception of neoplasms, deaths from these causes are more frequent among black children than white children. Additionally, infectious and parasitic diseases are also a major cause of death among black children.

^{19.} The results of this analysis may be found in Figures A3 and A4 and Tables A6 and A7.

^{20.} Earlier versions of this paper included additional covariates including age and calendar year dummies. We no longer include age dummies in our regression specification since these covariates are balanced on either side of the birth date threshold. In addition, we have replaced the calendar year dummies with the ICD-10 dummy in order to more directly control for changes in the classification of deaths over time.

^{21.} There are 18 chapters of internal causes in the ICD-10. Because deaths falling under a handful of chapters are extremely rare among children, we chose the 11 chapters with the highest number of deaths in our sample and grouped the remaining seven chapters into an "other" category for a total of 12 categories.

^{22.} We follow the List of 113 Selected Causes of Death (113-cause list) to create our subcategories for external causes. We have omitted deaths due to "Operations of war and their sequelae" since there were no deaths in our sample falling under this subcategory.

^{23.} Anderson et al. (2001) provide detailed information on the classification and rule changes between the ICD-9 and ICD-10 revisions and construct comparability ratios by case of death. They find substantial discontinuities in mortality trends resulting from implementing the ICD-10. Again, the inclusion of the dummy for the ICD-10 period in our regression specification should serve to control for this change.

the angle and a many column of a sume some some				la anna o		(aa.	Juno no	
		Black	Children			White	Children	
	4-7	8–14	15-18	19–23	4-7	8-14	15-18	19–23
Deaths from all causes	3.62	3.33	8.94	14.43	2.40	2.22	6.35	8.86
Deaths due to internal causes	1.53	1.46	2.32	3.71	1.20	0.99	1.39	2.02
Infections and parasitic diseases	0.15	0.15	0.17	0.37	0.08	0.06	0.07	0.08
Neoplasms (malignant and nonmalignant)	0.32	0.29	0.42	0.53	0.40	0.29	0.38	0.49
Diseases of blood, blood-forming organs, immune mechanism	0.07	0.06	0.07	0.19	0.02	0.02	0.02	0.02
Endocrine, nutritional, and metabolic diseases	0.05	0.05	0.10	0.22	0.07	0.07	0.10	0.15
Mental and behavioral disorders	0.02	0.01	0.03	0.06	0.02	0.01	0.03	0.09
Diseases of the nervous system	0.21	0.21	0.26	0.31	0.15	0.14	0.19	0.23
Cardiovascular and circulatory system diseases	0.14	0.20	0.47	0.80	0.09	0.11	0.21	0.32
Respiratory system diseases	0.15	0.21	0.25	0.28	0.09	0.08	0.11	0.13
Digestive system diseases	0.06	0.04	0.08	0.14	0.04	0.03	0.03	0.06
Congenital and chromosomal abnormalities	0.22	0.12	0.16	0.16	0.19	0.11	0.10	0.12
Symptoms and signs not elsewhere classified	0.09	0.08	0.18	0.30	0.04	0.04	0.11	0.24
All other diseases	0.03	0.05	0.14	0.35	0.03	0.02	0.04	0.09

 Table 2

 Average I

-
₽
0
E.
G
в
00
\mathbf{A}
q
Ц
а
05
ŭ
a
2
-
th
a
S
Ч
F
0
ŝe
m
a
\mathbf{O}
50
22
'n.
5
21
q
ž.
5
~
5
ŝ
∞
98
198
<i>J</i> , <i>198</i> .
30, 198.
r 30, 198.
er 30, 198
<i>iber 30, 198</i> .
mber 30, 198.
tember 30, 198.
ptember 30, 198.
September 30, 198.
September 30, 198.
re September 30, 198.
ore September 30, 1983
fore September 30, 1983
Before September 30, 1983
1 Before September 30, 1983
rn Before September 30, 1983
orn Before September 30, 1983
Born Before September 30, 1983
1 Born Before September 30, 1983
en Born Before September 30, 1983
lren Born Before September 30, 1983
ldren Born Before September 30, 1983
hildren Born Before September 30, 1983
Children Born Before September 30, 1983
Children Born Before September 30, 1983
or Children Born Before September 30, 1983
for Children Born Before September 30, 1983
h for Children Born Before September 30, 1983
ath for Children Born Before September 30, 1983
eath for Children Born Before September 30, 1983
Death for Children Born Before September 30, 1983
^c Death for Children Born Before September 30, 1983
of Death for Children Born Before September 30, 1983
2 of Death for Children Born Before September 30, 1983
tte of Death for Children Born Before September 30, 1983
Rate of Death for Children Born Before September 30, 1983
Rate of Death for Children Born Before September 30, 1983
ve Rate of Death for Children Born Before September 30, 198.
age Rate of Death for Children Born Before September 30, 1983

Deaths due to external causes	2.09	1.86	6.61	10.72	1.20	1.23	4.95	6.84
Accidents (unintentional injuries)	1.81	1.24	2.49	3.30	1.11	0.94	3.52	4.62
Transport accidents	0.83	0.63	1.79	2.55	0.62	0.63	2.94	3.35
Nontransport accidents	0.98	0.61	0.69	0.74	0.48	0.31	0.57	1.24
Falls	0.02	0.01	0.02	0.03	0.02	0.01	0.05	0.08
Discharge of firearms	0.05	0.08	0.15	0.11	0.02	0.06	0.07	0.05
Drowning and submersion	0.26	0.29	0.31	0.18	0.15	0.08	0.15	0.12
Exposure to smoke, fire, and flames	0.53	0.14	0.07	0.07	0.15	0.05	0.04	0.05
Poisoning/exposure to noxious substances	0.02	0.01	0.05	0.23	0.01	0.01	0.13	0.78
Other and unspecified	0.11	0.07	0.10	0.13	0.13	0.10	0.13	0.16
Intentional self-harm (suicide)	0.00	0.09	0.52	0.92	0.00	0.15	0.88	1.32
Assault (homicide)	0.23	0.49	3.47	6.33	0.07	0.11	0.48	0.69
Legal intervention	0.00	0.00	0.03	0.07	0.00	0.00	0.01	0.02
Events of undetermined intent	0.04	0.03	0.10	0.09	0.01	0.02	0.06	0.18
Operations of war and their sequelae	0:00	00:0	0:00	0.00	0.00	0:00	0:00	00:0
Complications of medical and surgical care	0.01	0.01	0.00	0.00	0.01	0.01	0.01	0.00

Notes: Average of annual rates of death per 10,000 children calculated for each birth month cohort between October 1979 and September 1983. Mortality rates constructed using National Vital Statistics System Multiple Cause of Death files for the years 1983. 2011 and NVSS Birth Data files for the years 1979–83.

A common cause of deaths due to external causes is transport-related accidents. Accidental drowning and submersion is also a frequent cause of death at younger ages. Among older children, deaths due to suicide and homicide are prevalent, with suicide more likely among white children and homicide more likely among black children.

IV. Results

Figure 3 graphs internal-cause mortality rates by birth month cohort for each race at ages four to seven, eight to 14, 15–18, and 19–23. In the panels on the left, we see very little evidence of a change in internal mortality for white children born after the September 30, 1983, cutoff. For black children, we see some evidence of a decline in internal mortality at the cutoff at ages eight to 14, although the plot is relatively noisy. There is particularly strong evidence, however, of a decline in internal mortality for black cohorts born after the cutoff at ages 15–18. This suggests a longer-term mortality improvement for black children under the expansions. There is no visual evidence, however, of a similar discontinuity at ages 19–23.

Table 3 provides the corresponding regression estimates. Each cell reports the estimated discontinuity for children at the September 30, 1983; birth date cutoff, or the coefficient β in Equation 1, from a regression for the specified age group and window size. Again, there is some evidence of a decline in internal mortality at ages eight to 14 for black children born after the cutoff—that is, during the period of extended Medicaid coverage. However, the coefficient estimates are only significant under the quadratic spline specification.

When we examine the longer-term impact of the expansions at ages 15-18, we find strong evidence supporting a decline in internal-cause deaths for black children. The coefficient estimates using the four-year observation window are statistically significant under all four functional form specifications. With a baseline rate of 2.32 deaths per 10,000, the point estimates indicate just over a 19 percent decrease (19.1–19.6) in the internal mortality rate for black children at these ages. For the most part, the coefficient estimates are robust to alternative window sizes, indicating similarly sized and significant declines. The estimates under the quadratic spline specification for the smaller windows are imprecise and in the case of the two-year window no longer negative. However, the confidence intervals on these estimates are much larger than under the other functional form specifications and do not allow us to rule out declines similar to those in the other specifications. Combined with the compelling visual evidence in Figure 3, these findings strongly indicate a mortality improvement for black children under the expansion in Medicaid eligibility. We do not, however, see evidence of similar declines at ages 19–23. Furthermore, neither the regression results nor the visual evidence indicate any improvement in the internal mortality of white children.

Figure 4 presents external-cause mortality rates by birth month cohort for each race and age group. The majority of panels in this figure reveal a strong downward trend in external mortality. Separate from this trend, we see little compelling evidence of changes in mortality for cohorts born after the September 30, 1983, threshold. If anything, the plots for white children suggest an increase in external mortality at the cutoff at ages eight to 14 and 15–18, as does the plot for black children at ages 15–18. We also see some evidence of a decline in external mortality at ages four to seven for black children, although this particular plot is fairly noisy.



Figure 3

Child Mortality from Internal Causes by Child Race

Notes: Mortality rates by cohort (per 10,000 children) were constructed using Multiple Cause of Death Data files for the years 1979–2011 and Birth Data files for 1979–87. Points represent means of the age-specific mortality rates for each birth cohort, as described in the text. The lines are fitted values from a regression that includes a quadratic in birth cohort and a dummy variable for children born after September 30, 1983.

(continued)



Figure 3 (continued)

Table 4 presents the regression estimates for external mortality. Again, we find some evidence of a baseline difference at ages four to seven for black children at the cutoff. However, the estimates are not consistently significant across functional forms and are not statistically significant when using smaller windows of observations. Without strong visual support as well, we do not interpret these estimates as convincing evidence of differences in external mortality for black children at these ages.

The evidence of an increase in external mortality for white children at the cutoff at ages eight to 14 is surprisingly more robust. Our estimates using the four-year observation window indicate a 7–9 percent increase in external mortality over the baseline of 1.23 deaths per 10,000 children. With the exception of the quadratic spline specification, the estimates using smaller windows around the cutoff are of similar size and statistically significant. Visually, however, Figure 4 does not present evidence of a change in external mortality of the size seen in the regression estimates.²⁴ The lack of compelling visual evidence in the raw data gives us pause in the interpretation of this finding as an effect of the Medicaid expansions. Later, we investigate the likelihood of encountering an estimate of this size by chance.

We find no evidence of a change in external mortality for either black or white children aged 15–18 and mixed evidence of a change for black children aged 19–23. The regression estimates under the two-year window and under the more flexible quadratic spline specification indicate a statistically significant decline in external mortality. However, the estimates using larger window sizes suggest much smaller declines and even reverse in sign. The plot of the raw data in Figure 4 does not indicate a discontinuity at the cutoff for this group. Taken together, we do not interpret this to be a robust finding.

^{24.} We should note that the inclusion of calendar month dummies in the regression specification appears to lead to slightly larger and more significant estimates. This suggests that seasonality might be making it difficult to discern a discontinuity in the graph. We should emphasize, however, that both the size and significance of the decline in internal mortality that we find for black cohorts born after September 30, 1983, are robust to the inclusion of calendar month dummies. Results for specifications without calendar month dummies may be found in Tables A8 and A9.

Table 3

Change in Annual Internal-Cause Mortality Rate for Children Born After September 30, 1983, by Race and Age Group

	В	lack Childre	n	W	hite Childi	ren
	Four-	Three-	Two-	Four-	Three-	Two-
	Year	Year	Year	Year	Year	Year
	Window	Window	Window	Window	Window	Window
Ages 4–7						
Linear	0.024	0.042	-0.009	0.036	-0.013	-0.031
	(0.089)	(0.104)	(0.132)	(0.041)	(0.049)	(0.056)
Linear spline	0.023	0.043	-0.006	0.036	-0.014	-0.030
	(0.090)	(0.103)	(0.129)	(0.041)	(0.049)	(0.056)
Quadratic	0.024	0.042	-0.009	0.036	-0.013	-0.031
	(0.089)	(0.104)	(0.134)	(0.041)	(0.048)	(0.056)
Quadratic spline	0.095	0.007	0.060	-0.005	-0.022	-0.037
	(0.114)	(0.129)	(0.155)	(0.061)	(0.070)	(0.074)
Baseline mean	1.525	1.562	1.560	1.205	1.186	1.192
Ν	192	144	96	192	144	96
Ages 8-14						
Linear	-0.098	-0.081	-0.144	0.014	0.018	0.029
	(0.090)	(0.099)	(0.125)	(0.029)	(0.033)	(0.042)
Linear spline	-0.099	-0.083	-0.146	0.013	0.017	0.028
	(0.086)	(0.096)	(0.122)	(0.028)	(0.032)	(0.041)
Quadratic	-0.094	-0.079	-0.142	0.014	0.018	0.029
	(0.087)	(0.097)	(0.123)	(0.028)	(0.032)	(0.041)
Quadratic spline	-0.206*	-0.268**	-0.321**	0.012	-0.001	-0.045
	(0.109)	(0.117)	(0.153)	(0.038)	(0.041)	(0.050)
Baseline mean	1.462	1.489	1.484	0.988	0.980	0.971
Ν	288	216	144	288	216	144
Ages 15–18	-0.443***	-0.465***	-0.460**	0.028	-0.000	-0.009
Linear	(0.126)	(0.151)	(0.195)	(0.048)	(0.052)	(0.068)
Linear spline	-0.447***	-0.465***	-0.453**	0.025	-0.000	-0.006
	(0.124)	(0.148)	(0.195)	(0.045)	(0.050)	(0.065)
Quadratic	-0.448***	-0.471***	-0.467**	0.022	-0.001	-0.010
	(0.125)	(0.148)	(0.194)	(0.046)	(0.051)	(0.066)

(continued)

Table 3 (continued)

	В	lack Childre	n	W	hite Child	ren
	Four-	Three-	Two-	Four-	Three-	Two-
	Year	Year	Year	Year	Year	Year
	Window	Window	Window	Window	Window	Window
Quadratic spline	-0.456**	-0.349	0.053	0.008	0.011	0.077
	(0.197)	(0.258)	(0.320)	(0.071)	(0.072)	(0.085)
Baseline mean	2.322	2.387	2.440	1.393	1.378	1.384
Ν	192	144	96	192	144	96
Ages 19–23						
Linear	0.052	0.098	0.067	-0.007	-0.024	-0.071
	(0.127)	(0.143)	(0.162)	(0.054)	(0.065)	(0.084)
Linear spline	0.049	0.096	0.070	-0.009	-0.024	-0.070
	(0.126)	(0.143)	(0.164)	(0.054)	(0.065)	(0.085)
Quadratic	0.052	0.098	0.067	-0.007	-0.024	-0.071
	(0.125)	(0.143)	(0.163)	(0.054)	(0.065)	(0.085)
Quadratic spline	0.120	0.091	0.175	0.005	-0.041	-0.045
	(0.174)	(0.189)	(0.234)	(0.070)	(0.086)	(0.097)
Baseline mean	3.712	3.694	3.680	2.019	2.018	2.011
Ν	192	144	96	192	144	96

Notes: Coefficients from OLS regressions with the specified function in birth cohort, a dummy for deaths classified under ICD-10 and calendar month dummies. Standard errors reported in parentheses are robust and clustered by birth month cohort, *** p < 0.01, ** p < 0.05, * p < 0.1.

A. Specific Causes of Death

Our results provide compelling evidence of a decline in mortality under the expansions for black children born after September 30, 1983, aged 15–18. We also find evidence suggesting a smaller increase in external mortality for white children born after the cutoff at ages eight to 14. To better understand the potential role of expanded Medicaid eligibility behind these changes in mortality, we study more specific underlying causes of death at these ages using the classification described earlier. The existing literature examining Medicaid mortality effects has not examined changes in specific cause mortality.

Since total mortality rates for children are fortunately low, deaths due to specific causes are particularly low frequency events. To better model the determinants of specific causes of death, we focus on Poisson regression estimates. Nevertheless, in our regression discontinuity framework that works best with a very large sample, our estimates still rely heavily on sometimes very noisy observations for the months immediately surrounding the September 1983 cutoff. For this reason, although we report the estimated



Figure 4



Notes: Mortality rates by cohort (per 10,000 children) were constructed using Multiple Cause of Death Data files for the years 1979–2011 and Birth Data files for 1979–87. Points represent means of the age-specific mortality rates for each birth cohort, as described in the text. The lines are fitted values from a regression that includes a quadratic in birth cohort and a dummy variable for children born after September 30, 1983.

(continued)



Figure 4 (continued)

marginal effects from this analysis in the Appendix, we do not emphasize the point estimates themselves in the discussion that follows.

We begin by examining changes in specific causes of internal mortality for black children aged 15–18. The results suggest that the largest contributors to the decline in internal mortality are decreases in deaths due to diseases of the nervous system and to neoplasms, although the estimates vary in size and significance across specifications and observation windows (estimates are reported in Table A10). We also see declines for black children spread across other causes, including infectious and parasitic diseases, mental and behavioral disorders, cardiovascular and circulatory system diseases, and respiratory system diseases. Although individually insignificant, these estimates in the aggregate support a large decline in the total number of internal-cause deaths occurring among older black children.

Next, we examine changes in specific causes of external mortality for white children at ages eight to 14 (estimates reported in Table A11). Again, the estimates in this analysis vary in size and significance across specifications and observation windows. But the evidence largely suggests an increase in deaths resulting from transportation-related accidents, accidental poisoning and exposure to noxious substances, and "other" nontransportation accidents. The top causes for this latter category at these ages include accidental mechanical suffocation, accidents caused by machinery, and accidents caused by electric currents. There is also evidence of an increase in deaths resulting from intentional self-harm or suicide.

B. Specification Checks

In the results reported above, the finding of a decrease in internal mortality among black cohorts gaining Medicaid at ages 15–18 holds up to a range of functional form specifications and window sizes. It also passes visual inspection. We also find persistent evidence, albeit to a lesser extent, of an increase in external mortality among white cohorts gaining Medicaid eligibility during the period of coverage gain.

To gauge how likely it would be to arrive at similar effect sizes by chance, we estimate jumps at nondiscontinuity points for these outcomes and age ranges. We run a series of

Table 4

Change in Annual External-Cause Mortality Rate for Children Born After September 30, 1983, by Race and Age Group

	I	Black Childre	n	V	White Childre	n
	Four-Year	Three-Year	Two-Year	Four-Year	Three-Year	Two-Year
	Window	Window	Window	Window	Window	Window
Ages 4–7						
Linear	-0.269*	-0.254	-0.257	-0.074	-0.010	0.022
	(0.136)	(0.169)	(0.193)	(0.045)	(0.047)	(0.058)
Linear spline	-0.275**	-0.267*	-0.276	-0.075	-0.010	0.022
	(0.122)	(0.137)	(0.167)	(0.045)	(0.048)	(0.058)
Quadratic	-0.269**	-0.254*	-0.257	-0.074	-0.010	0.022
	(0.128)	(0.141)	(0.165)	(0.045)	(0.047)	(0.058)
Quadratic spline	-0.226	-0.108	-0.191	0.086	0.088	0.075
	(0.160)	(0.192)	(0.244)	(0.065)	(0.072)	(0.084)
Baseline mean	2.093	2.094	2.147	1.196	1.197	1.174
Ν	192	144	96	192	144	96
Ages 8-14						
Linear	-0.048	-0.127	-0.098	0.107**	0.105**	0.123*
	(0.096)	(0.112)	(0.134)	(0.041)	(0.050)	(0.063)
Linear spline	-0.049	-0.130	-0.100	0.106***	0.103***	0.118**
	(0.094)	(0.111)	(0.136)	(0.031)	(0.038)	(0.046)
Quadratic	-0.046	-0.123	-0.095	0.110***	0.108***	0.128***
	(0.094)	(0.109)	(0.135)	(0.033)	(0.039)	(0.047)
Quadratic spline	-0.119	-0.054	-0.120	0.089*	0.071	0.032
	(0.148)	(0.157)	(0.164)	(0.046)	(0.054)	(0.056)
Baseline mean	1.864	1.792	1.785	1.227	1.183	1.146
Ν	288	216	144	288	216	144
Ages 15-18						
Linear	0.185	0.025	-0.072	0.162	0.035	0.006
	(0.423)	(0.453)	(0.488)	(0.203)	(0.206)	(0.170)
Linear spline	0.101	0.023	-0.039	0.118	0.033	0.022
	(0.247)	(0.294)	(0.390)	(0.095)	(0.110)	(0.121)
Quadratic	0.041	-0.042	-0.094	0.085	-0.002	-0.006
	(0.278)	(0.320)	(0.421)	(0.101)	(0.107)	(0.117)

(continued)

Table 4 (continued)

	E	Black Childre	n		White Childre	n
	Four-Year	Three-Year	Two-Year	Four-Year	Three-Year	Two-Year
	Window	Window	Window	Window	Window	Window
Quadratic spline	-0.062	-0.251	-0.312	0.055	0.053	0.030
	(0.307)	(0.349)	(0.449)	(0.135)	(0.160)	(0.145)
Baseline mean	6.613	6.249	5.951	4.953	4.812	4.693
Ν	192	144	96	192	144	96
Ages 19–23	0.071	-0.294	-0.993***	0.018	-0.074	-0.115
Linear	(0.260)	(0.275)	(0.331)	(0.120)	(0.121)	(0.139)
Linear spline	0.067	-0.312	-1.024***	0.006	-0.085	-0.121
	(0.256)	(0.239)	(0.302)	(0.098)	(0.111)	(0.139)
Quadratic	0.071	-0.294	-0.993***	0.018	-0.074	-0.115
	(0.260)	(0.242)	(0.296)	(0.094)	(0.107)	(0.139)
Quadratic spline	-0.854***	-1.305***	-1.134***	-0.163	-0.246*	-0.211
	(0.304)	(0.304)	(0.404)	(0.131)	(0.136)	(0.175)
Baseline mean	10.723	10.522	10.488	6.840	6.889	6.978
Ν	192	144	96	192	144	96

Notes: Coefficients from OLS regressions with the specified function in birth cohort, a dummy for deaths classified under ICD-10 and calendar month dummies. Standard errors reported in parentheses are robust and clustered by birth month cohort, *** p < 0.01, ** p < 0.05, * p < 0.1.

placebo tests for breaks occurring either prior to or following the September 30, 1983, cutoff. To examine jumps in external mortality for whites aged eight to 14, we estimate breaks at the 120 available cutoff points using a rolling four-year window of data among cohorts born either entirely before or after the cutoff.²⁵ Each regression includes a quadratic function in birth cohort, a dummy for deaths classified under ICD-10, and calendar month dummies. The top panel of Figure A5 presents the resulting RD estimates for all 120 simulations. Ten, or 8 percent, of the simulations lead to an estimate larger in absolute value than our estimate at the September 30, 1983, cutoff.

We run a similar series of simulations to estimate breaks in internal mortality at nondiscontinuity points for blacks at ages 15-18.²⁶ The bottom panel of Figure A5

^{25.} We estimate breaks at each month of birth between January 1975–September 1979 and October 1987–December 1988. For this exercise, we are using mortality rates that are constructed by dividing the number of deaths by the unadjusted birth cohort size. We are unable to subtract out deaths prior to the age in question for cohorts born prior to 1979 since birth date information is only available in the mortality data files starting in this year.
26. The data available allows us to estimate breaks at each month of birth between January 1972–September

^{26.} The data available allows us to estimate breaks at each month of birth between January 1972–September 1979 and October 1987–December 1992.

presents the distribution of coefficient estimates. In a total of 108 simulations, zero (0 percent) of the simulations lead to an estimate larger in absolute value than the estimated decline in the internal mortality rate at September 30, 1983. This analysis strongly suggests that our estimated decline in black internal mortality is larger than estimates that we might observe due to chance. In contrast, it appears more likely that we might be picking up an estimate for whites that may not necessarily be linked to the change in Medicaid policy.

This agrees with findings from another check on our results indicating the absence of a robust relationship between Medicaid eligibility and white external mortality (included in the working paper Meyer and Wherry 2012). When calculating simple differences that compare the mortality of children born during the 12 months before and after the September 30, 1983, cutoff, we find no significant difference in the external mortality rates of white children born before and after the cutoff. In contrast, we find significant evidence of a decline in internal mortality for black children at ages 15–18. A difference-in-differences estimate that compares this change to that experienced by white children indicates a significant decline of 0.37 deaths per 10,000 black children.²⁷

One potential concern with the RD design is the possibility of other changes affecting birth cohorts at the September 30, 1983; cutoff. If these changes affect the health of these cohorts, we may erroneously attribute differences in later-life mortality to the gain in Medicaid eligibility. To investigate this possibility, we test for breaks in cohort characteristics that existed prior to the change in Medicaid policy. Using natality data available for the birth cohorts in our sample, we construct and test measures of cohort size, the mean age of the mother, the shares of each cohort with a high school-educated mother and married mother, receipt of prenatal care at any time during the pregnancy, and receipt of prenatal care in the first trimester. We also test for differences in early life health by examining measures of the prevalence of preterm birth, low birth weight, and very low birth weight births for each cohort.

Figures A6 and A7 present plots for each of these cohort characteristics by child race. Regression analyses (reported in Tables A12 and A13) indicate some evidence of a decrease in cohort size, a decline in marriage among mothers, and a small increase in receipt of any prenatal care (one-fifth of a percentage point over a baseline of 97 percent) among black cohorts born after the September 30, 1983, cutoff. The size and statistical significance of the covariates, however, vary across specifications and window sizes. Among white cohorts, we find stronger evidence of an increase in the receipt of any prenatal care for cohorts born after the cutoff, although again the change is small in size (one-tenth of a percentage point over a base of 99 percent). Importantly, for both race groups, we find no evidence of differences in infant health at the September 30, 1983, cutoff.

Given the sensitivity of the estimates that are significant, we do not view the results of these tests as a concern for the interpretation of our findings. The evidence of a change in birth cohort size at the cutoff for blacks is not particularly robust and is accounted for in

^{27.} Results may be found in Table 8 in Meyer and Wherry (2012). While providing a simple plausibility check on our RD results, this analysis relies on several strong assumptions. The differences estimates assume that there are no trends in mortality rates, while the difference-in-differences assume that any underlying trends in mortality are similar across race groups. In addition, the difference-in-differences estimate should be interpreted as the relative difference in the effect of the expansions for black children compared to white children, acknowledging that white children born after the cutoff also benefitted from the Medicaid expansions.

our construction of mortality rates. And without the concern of the potential manipulation of birth dates, continuity of the density of births at the cutoff is not necessary for the RD design (Imbens and Lemeiux 2008). The observed changes in mother's marital status and prenatal care receipt are small and, if they were to represent actual discontinuities, are unlikely to account for changes in later-life health of the magnitude we find in the mortality data. When we include these variables in our RD specification, the estimates of the decline in internal mortality for blacks are robust in their magnitude, although we do lose some precision (Table A14). Finally, placebo simulations that test for breaks in these characteristics at nondiscontinuity points indicate that it is relatively common to observe jumps of a similar magnitude at other cutoffs.²⁸

C. Additional Analyses

In addition to the analyses already presented, we also explored whether changes in mortality differed among children residing in states with more or less generous expansions in Medicaid eligibility. Across states, there is variation in the average gain in years of public health insurance eligibility at the September 30, 1983; cutoff due to differences in Medicaid policies before and after the expansions.²⁹ We should emphasize that this source of variation in Medicaid eligibility across states is not as clearly exogenous as birth date. States with improving child mortality may be less likely to expand coverage, for example. However, this type of variation is the main source of identifying variation that has been used in the literature on Medicaid expansions.

To estimate the eligibility gain for each state, we follow the literature and use a simulated eligibility measure constructed using the national sample of children. This captures variation in average eligibility gains by state resulting from legislative rather than sociodemographic differences. Ranking the state estimates, we divide the states into two groups. The "high-eligibility" group includes the states with the largest average gain in years of eligibility, while the "low-eligibility" group includes the states with the smallest average gain in years of eligibility.³⁰ We repeat the analysis for each state group using state of residence information available in the mortality records.³¹

^{28.} See Appendix C for additional details on this analysis and the placebo simulations.

^{29.} States' eligibility criteria in place prior to the expansions depended on state-determined AFDC program rules and optional eligibility programs. In addition, state decisions regarding optional Medicaid and CHIP expansions varied in terms of timing and child age and family incomes covered.

^{30.} We use this method to identify the high- and low-eligibility gain states for children of each race in addition to the identification for all children. Table A15 reports the state groups for all children and for each race group. Table A16 presents summary characteristics for children in each of the state groups in 1983 prior to the first Medicaid eligibility expansions. The table suggests that more children stood to gain from the Medicaid eligibility expansions in the high eligibility gain states due to lower preexpansion eligibility levels for Medicaid.

^{31.} To construct denominators in this analysis, we rely on birth cohort size by state. The mobility of children and their families between state groups before, during, or after the period of eligibility gain may influence our estimates. When we conduct the analysis using the logged number of deaths rather than rates of death, we have similar findings. Mobility is still a concern, however, if children move across state groups from the period of coverage to later ages. Encouragingly, during this time period children were less likely to move to a different state at ages ten to 17 than at earlier ages (U.S. Census Bureau 2001). The estimates indicate that less than 3 percent of children of these ages move to a different state per year. In addition, a sizeable share of these movers would presumably relocate to states that are members of the same state group.

We do not find major differences in the mortality effects across state groups.³² In particular, we find no evidence that children residing in the high-eligibility gain states saw greater improvements in health. When we examine the results for each state group by child race, we find evidence of a decline in internal mortality for older black children in both groups. However, if anything, our results are indicative of a slightly larger improvement in black mortality in the low-eligibility gain states.

We find a similarly counterintuitive finding in a subsequent analysis that divides deaths into counties with high and low child poverty rates during the period of expansion.³³ We find evidence of a significant decrease in internal mortality at ages 15–18 among all children in counties with lower poverty rates and no similar improvement in counties with high poverty rates. When we examine changes in mortality in each group by child race, we find evidence of declines in internal mortality among blacks in both county groups. However, our coefficient estimates suggest a larger improvement in black mortality in the low-poverty counties.

One potential explanation for not finding stronger improvements in health in the expected locations in these analyses is that larger expansions in eligibility may not directly translate into higher levels of enrollment or improvements in access to medical care for children. Increases in enrollment require active outreach by states and localities and the dissemination of eligibility information to parents. In addition, larger numbers of new beneficiaries may strain limited supply-side resources, particularly if physicians opt not to accept Medicaid clients due to low reimbursement rates or slow payment.

V. Interpreting the Estimates: Plausibility, Comparisons to Past Work and Cost Per Life Saved

In this paper, we find evidence of a sizeable decrease in the internal mortality rate of older black teens as a result of increased childhood eligibility for Medicaid. Our point estimate indicates a 19 percent decrease at ages 15–18. We find no evidence of a similar effect among white children, although the smallest of our implied 95 percent confidence intervals allows us only to rule out a decline greater than 4.5 percent for white teens. This finding is consistent with previous work examining the Medicaid expansions of this period. In their working paper, Currie and Gruber (1995) estimate that an increase in the fraction of children eligible for Medicaid at the state level had a much larger impact (more than four times as big) on mortality for blacks than for whites, although neither estimate is significantly different from zero. The authors also

^{32.} Results for all races may be found in Figures A9 and A10 and Tables A17 and A18. Results for each state group by child race are available from the authors.

^{33.} We classify counties into two groups using the population-weighted median of 1989 poverty rates by county for children aged zero to 17 from the Census Small Area Income and Poverty Estimates as a cutoff. Using information on county of residence in the mortality data, we then total deaths in the "high-poverty" and "low-poverty" counties and rerun the analysis for each group using the log number of deaths as the dependent variable. The results for all races are reported in Tables A19 and A20 and Figures A11 and A12. Results for each county group by child race are available from the authors.

provide suggestive evidence that newly covered black children under the Medicaid expansions used more healthcare than their white counterparts. As noted below, our overall effect of the eligibility expansions on mortality is smaller than found in Currie and Gruber (1996a).

The size of our point estimate for black children is large but more reasonable when we consider that deaths among black children are highly concentrated among those who would gain coverage under the expansions. Using information on deaths for children aged 15–18 prior to the expansions in the 1993 National Mortality Followback Study, we estimate that 64 percent of the deaths occurring among black children at these ages were to decedents with household incomes below the poverty line.³⁴ Combining this information with our measured 19 percent decline in internal mortality among blacks under the expansions, this implies a 30 percent (19 divided by 0.64) decline in the internal mortality rate for poor black children.³⁵ The tightest of the confidence intervals on our estimates for black teens indicate that the eligibility effect is at least a 9 percent decline in internal mortality with 95 percent probability (or a decline of 0.20 deaths per 10,000). This lower-bound estimate would imply a 12.5 percent decline for poor black children.

This estimate is large but not implausible given the large increases in eligibility for poor black children. Our first-stage estimates indicate that 32 percent of poor black children at the birth date cutoff gained an average of 5.2 years of Medicaid eligibility under the expansions (not shown in Table 1). In reality, it is more likely that a larger share of children gained eligibility but for shorter intervals of time given fluctuations in income and known "churn" in Medicaid enrollment for children (see Fairbrother et al. 2007). And, while existing evidence indicates low takeup rates of Medicaid coverage (for instance, 8 percent as estimated by Card and Shore-Sheppard 2004), a new study of Medicaid enrollment under these expansions by race indicates that takeup for black children was higher than average at approximately 30 percent (Wherry et al. 2015). The authors find no evidence of a change in Medicaid coverage for nonblack children. For the children who did enroll in Medicaid, sustained increases in insurance coverage could result in cumulative effects that are larger in magnitude than previously measured contemporaneous effects.

Finally, it is reasonable to expect that children who do take up coverage are those most in need of medical care (specifically, the conditional-coverage phenomenon; see Marton and Yelowitz 2014).³⁶ Our analysis of specific causes indicates that the deaths that were avoided were most likely related to serious and chronic childhood disease, which would support this hypothesis. It is worth noting also that even if the

^{34.} Approximately 40 percent of black children aged 15-18 lived in poverty in 1993 (authors' calculation from the March supplement to the 1994 Current Population Survey), indicating that the baseline death rate for poor black children was 0.64/0.4 = 1.6 times higher than for the general black population.

^{35.} This calculation assumes that the change in deaths estimated under the policy occurred among poor children only. It also assumes that the share of internal-cause deaths for black children aged 15–18 with incomes under poverty is similar to the share calculated for all deaths in the National Mortality Followback Study.

^{36.} In their study of takeup and the utilization of health care under OBRA90, Card and Shore-Sheppard (2004) find evidence that supports this hypothesis. They estimate that children with new Medicaid coverage are 60 percent more likely to have an annual doctor visit than those without coverage.

typical newly eligible child is not enrolled in Medicaid, upon commencing any costly hospital or physician care, it is likely that an eligible child would be enrolled or receive retroactive benefits.³⁷

The decrease in later-life mortality for black children is supported by recent work documenting a sizeable decline in adult hospitalizations and emergency department (ED) visits among the cohorts gaining childhood Medicaid eligibility. Using a similar regression discontinuity model, the point estimates from Wherry et al. (2015) indicate that black children born after the September 30, 1983, cutoff experienced between 8 and 13 percent fewer hospitalizations and 3–4 percent fewer ED visits at age 25 relative to those born just before the cutoff. The effects of child Medicaid eligibility are even more pronounced for hospital and ED visits related to chronic health conditions, which are plausibly influenced by access to care in childhood.

Surprisingly, we also find some evidence of an increase in external mortality for white children at the cutoff at ages eight to 14. The point estimates from the regressions suggest between a 7 and 9 percent increase in external mortality. As discussed earlier, the evidence for this finding is less convincing than the decrease we observe in internal mortality for blacks. Sensitivity analyses indicate that there is a reasonable probability that an estimate of this magnitude may be found due to chance. Combined with less-than-persuasive graphical evidence, we are wary of interpreting it as an effect of the Medicaid expansions. That said, the change occurs during the period of eligibility gain and possible explanations could include moral hazard or income effects that lead to increases in risky behaviors (see the discussion in Cutler and Zeckhauser 2000 and Cawley and Ruhm 2011, respectively), reductions in healthcare quality under the crowdout of private coverage, or the consequence of strained supply-side resources under expansions in overall levels of coverage.

A. Comparisons to Previous Work

The results presented here diverge from the literature in two important ways. First, we are unable to detect significant evidence of a contemporaneous decline in child mortality associated with expanded childhood eligibility for public health insurance. We did, however, find persuasive evidence of longer-term mortality effects among black children during the period just following the gain in public coverage. This is the first study documenting the potential longer-term mortality effects of expansions in public health insurance among children.³⁸

Second, our estimate for the mortality effect of public eligibility is smaller than the estimate found by Currie and Gruber (1996a). We use some rough calculations to demonstrate this point. In our four-year window analysis, we estimate a decline in the annual internal mortality rate of older black children aged 15–18 of about 0.44 deaths per 10,000 children. With black children representing approximately 15.6 percent of all

^{37.} Eligibility could be granted retroactively for up to three months prior to the date of application. In addition, many states were giving children the opportunity to apply for Medicaid at the sites where they received health care (Congressional Research Service 1993).

^{38.} A new working paper by Brown, Kowalski, and Lurie (2015) examines the impact of Medicaid eligibility during childhood on adult mortality and economic outcomes, using IRS data. The authors find evidence of mortality decreases for adults in their late twenties; they do not study mortality during the adolescent years.

children and our calculation of 0.42 years of eligibility gain on average for all children, our estimates imply an overall decline of $(0.44 \times 4 \times 0.156) \div 0.42 = 0.654$ deaths per 10,000 children of all races gaining one year of eligibility.³⁹ This mortality derivative with respect to eligibility ignores any potential adverse effect for white children and yet remains about half the size of that found by Currie and Gruber. They found that 1.277 deaths in a year were avoided for every 10,000 children aged one to 14 of all races made eligible for that year.⁴⁰

B. Cost Per Life Saved

To further interpret our results, we estimate the approximate cost to the Medicaid program of each child life saved.⁴¹ By its nature, this calculation ignores any effects of the expansions on morbidity or quality of life. To calculate the cost of provision of Medicaid services for every 10,000 children, we again use the estimate of an 8 percent takeup rate under the expansions. With the average gain in eligibility of 0.42 years among all children, this rate implies an increase in Medicaid coverage of $0.42 \times 0.08 = 0.034$ years per child. Using an estimate of the cost of Medicaid coverage per child during this time period, this implies the Medicaid expansion cost \$1,430 $\times 0.034 \times 10,000 = $486,200$ (in 2011 dollars) for every 10,000 children.⁴² With our estimate that $0.44 \times 4 \times 0.156 = 0.275$ deaths per 10,000 children were prevented by the expansions, we calculate that it cost \$486,200 $\div 0.275 = 1.77 million (in 2011 dollars) to save one life.⁴³

Although these calculations are similar to those used in the existing literature, this estimate may understate costs since it relies on the average cost of Medicaid coverage rather than the marginal cost for newly enrolled children. Using data from the Medicaid Statistical Information System, Brown, Kowalski, and Lurie (2015) estimate that each year of additional Medicaid eligibility under the expansions increased Medicaid spending by \$575 in 2011 dollars. This would imply an increase in Medicaid spending of \$2.42 million per 10,000 children made eligible and therefore a cost per life saved of \$8.8 million. Although this indicates a smaller return, when combined with other potential long-term benefits of public health insurance including improvements in economic outcomes (Brown, Kowalski, and Lurie 2015; Cohodes et al. 2014) and reductions in healthcare utilization (Miller and Wherry 2014, Wherry et al. 2015), there may be significant offsets to the cost of these expansions.

^{39.} In 2000, 15.6 percent of all children in the United States were black. We consider only the reduction to black mortality observed at ages 15–18 in our calculation and assume that the expansions had no effect on mortality at other ages or for other race groups. We multiply 0.44 by four for the four years between ages 15–18.
40. While the literature does not provide a comparable estimate, when we consider a takeup rate among newly eligible children of 8 percent, as estimated by Card and Shore-Sheppard (2004), our estimates imply a decrease of 8.175 deaths per year for every 10,000 children newly enrolled in Medicaid under the expansions.

^{41.} The back-of-the-envelope calculations in this section implicitly assume a social discount rate of zero. A higher rate of social discount would imply a somewhat higher cost per life saved.

^{42.} The per capita Medicaid payment for categorically needy AFDC children in FY 1992 was \$891 (U.S. House of Representatives 1996). This is \$1,430 in 2011 dollars.

^{43.} Despite the differences in our estimated mortality effects, this estimate is lower than the \$1.61 million (1992 dollars) cost per child life saved (\$2.68 in 2011 dollars) calculated by Currie and Gruber (1996a). This difference is due mostly to the lower estimate of the takeup of public health insurance in our context.

VI. Conclusions

The effectiveness of public health insurance in improving the health of children is an important policy question. Despite drastic expansions in public health insurance over the last 30 years, the evidence connecting this expanded coverage to improved child health remains hard to come by. Identifying the impact of health insurance on health is a well-known problem since individuals with insurance differ from those without and since health itself may affect coverage (Levy and Meltzer 2008). While randomly assigning health insurance overcomes this problem, there have been few opportunities for experimental study. Only one experimental study has compared individuals with and without health insurance coverage, and its focus was on low-income adults not children.

This paper uses a novel quasi-experimental design to provide new evidence on the health effects of public health insurance for children. While not an experimental study, our research design approximates random assignment in public health insurance for children by exploiting changes in Medicaid eligibility rules that differentially affected those born before and after September 30, 1983. Early expansions under Medicaid in the 1980s and 1990s extended eligibility only to children born after this date. We show this led to a large, cumulative difference in public health insurance eligibility for children born on either side of the September 30, 1983; cutoff. Children in families with incomes at or just below the poverty line gained close to five additional years of eligibility if they were born in October 1983 rather than just one month before.

An additional advantage of the research design is that variation in childhood eligibility is tied to birth date, which provides an easy way to track individuals over time to study the longer-term effects of insurance coverage. In general, very little is known about the long-term effects of health insurance coverage on health (Cutler and Zeckhauser 2000). It is possible that the payoffs from public health insurance coverage, particularly for children, might not be fully evident until later in life. Without considering the long-term effects of insurance coverage, we may be missing an important component of program benefits.

Using a regression discontinuity design, we examine the impact of public health insurance eligibility during childhood on immediate and longer-term mortality by comparing outcomes for cohorts of children born just before and after the September 30, 1983, cutoff. The discontinuity in public health insurance eligibility under the expansions occurred between the ages of eight and 14. We measure mortality outcomes for this period, as well as through age 23, to gauge any longer-term impacts. We examine differential effects of the expansions by the race of the child because black children were more likely to be affected by the Medicaid expansions. They gained more than twice the years of eligibility on average as white children at the September 30, 1983, birth date cutoff.

The results provide compelling evidence of an improvement in mortality for black children under the Medicaid expansions. Although there is some suggestion of a decline in mortality rates during the period of coverage (ages eight to 14), we find strong support for a sizeable decline in the later-life internal-cause mortality of black children aged 15–18. With deaths due to internal causes likely to be influenced by access to medical care, this result supports an improvement in the underlying health of black children under the

eligibility expansions. Furthermore, we find evidence that this gain in health is not reversed during the early adult years.

We find no evidence of a similar decline in the mortality of white children under the expansions. Surprisingly, we find some evidence of an increase in external mortality for white children during the period of coverage gain. We mention several potential explanations for this result but also provide evidence that keeps us from drawing any definitive conclusion. Future work may be able to resolve whether there are adverse effects of public expansions for certain population groups.

This paper provides important new evidence of longer-term mortality improvements among black children under public health insurance. There are several limitations to our analysis and to the generalizability of our findings. First, as is the case with all quasiexperimental designs, the identification of program effects is susceptible to confounders that, in this case, would differentially affect cohorts born on either side of the September 30, 1983, birth date cutoff. However, we know of no other social policy or economic influences during this time period that would have differentially affected cohorts born before and after this date. Second, our estimates rely on variation in Medicaid coverage for children in the 1990s, which may be limited in generalizability to contemporary changes in Medicaid coverage or for different populations gaining coverage. Third, although we investigate changes in the underlying causes of death, we are limited in our ability to shed more light on the specific mechanisms behind the observed changes in mortality. Finally, our analysis is limited by the amount of time that has passed following the policy change. With time (and further study), we will be able to look at longer-term adult outcomes for the cohorts of children affected by the change in Medicaid policy.

References

- Anderson, Robert N., Arialdi M. Minino, Donna L. Hoyert, and Harry M. Rosenberg. 2001. "Comparability of Cause of Death Between ICD-9 and ICD-10: Preliminary Estimates." *National Vital Statistics Reports* 49(2):1–32.
- Baicker, Katherine, Sarah L. Taubman, Heidi L. Allen, Mira Bernstein, Jonathan H. Gruber, Joseph P. Newhouse, Eric C. Schneider, Bill J. Wright, Alan M. Zaslavsky, and Amy N. Finkelstein. 2013. "The Oregon Experiment—Effects of Medicaid on Clinical Outcomes." New England Journal of Medicine 368(18):1713–22.
- Becker, Gary. 2007. "Health as Human Capital: Synthesis and Extensions." *Oxford Economic Papers* 59(3):379–410.
- Boudreaux, Michel H., Ezra Golberstein, and Donna McAlpine. 2014, "The Long-Term Impacts of Medicaid Exposure in Early Childhood: Evidence from the Program's Origin." Working paper,
- Brown, David, Amanda E. Kowalski, and Ithai Z. Lurie. 2015. "Medicaid as an Investment in Children: What Is the Long-Term Impact on Tax Receipts?" NBER Working Paper 20835.
- Buckles, Kasey, and Daniel M. Hungerman. 2013. "Season of Birth and Later Outcomes: Old Questions, New Answers." *Review of Economics and Statistics* 95(3):711–24.
- Card, David, and Lara Shore-Sheppard. 2004. "Using Discontinuous Eligibility Rules to Identify the Effects of the Federal Medicaid Expansions." *Review of Economics and Statistics* 86 (3):752–66.
- Case, Anne, Darren Lubotsky, and Christina Paxson. 2002. "Economic Status and Health in Childhood: The Origins of the Gradient." The American Economic Review 92(5):1308–34.

Cassedy, Amy, Gerry Fairbrother, and Paul W. Newacheck. 2008. "The Impact of Insurance Instability on Children's Access, Utilization, and Satisfaction with Health Care." Ambulatory Pediatrics 8(5):321–28.

Cawley, John, and Christopher J. Ruhm. 2011. "The Economics of Risky Health Behaviors." In *Handbook of Health Economics* Volume 2, ed. Mark V. Pauly, Thomas G. Mcguire, and Pedro P. Barros, 95–199. Oxford: Elsevier.

Cohodes, Sarah, Daniel Grossman, Samuel Kleiner, and Michael F. Lovenheim. 2014, "The Effect of Child Health Insurance Access on Schooling: Evidence from Public Insurance Expansions." NBER Working Paper 20178,

Congressional Research Service. 1993. "Medicaid Source Book: Background Data and Analysis (A 1993 Update)." A Report prepared by the Education and Public Welfare Division of the Congressional Research Service for the use of the Subcommittee on Health and the Environment of the Committee on Energy and Commerce, U.S. House of Representatives.

Currie, Janet, and Jonathan Gruber. 1995. "Health Insurance Eligibility, Utilization of Medical Care, and Child Health." NBER Working Paper 5052.

——. 1996a. "Health Insurance Eligibility, Utilization of Medical Care, and Child Health." *Quarterly Journal of Economics* 111(2):431–66.

———. 1996b. "Saving Babies: The Efficacy and Cost of Recent Changes in the Medicaid Eligibility of Pregnant Women." *Journal of Political Economy* 104(6):1263–96.

Currie, Janet, Sandra Decker, and Wanchuan Lin. 2008. "Has Public Health Insurance for Older Children Reduced Disparities in Access to Care and Health Outcomes?" *Journal of Health Economics* 27:1567–81.

- Cutler, David M., and Jonathan Gruber. 1996. "Does Public Insurance Crowd Out Private Insurance?" *Quarterly Journal of Economics* 111(2):391–430.
- Cutler, D.M., and R.J. Zeckhauser. 2000. "The Anatomy of Health Insurance." In *Handbook of Health Economics* Volume 1A, ed. Anthony J. Culyer and Joseph P. Newhouse, 563–643. Oxford: Elsevier.

Dafny, Leemore, and Jonathan Gruber. 2005. "Public Insurance and Child Hospitalizations: Access and Efficiency effects." *Journal of Public Economics* 89:109–29.

De la Mata, Dolores. 2012. "The Effect of Medicaid Eligibility on Coverage, Utilization, and Children's Health." *Health Economics* 21(9):1061–79.

Fairbrother, Gerry Lynn, Heidi Park Emerson, and Lee Partridge. 2007. "How Stable Is Medicaid Coverage for Children?" *Health Affairs* 26(2):520–28.

- Finkelstein, Amy, Sarah Taubman, Bill Wright, Mira Bernstein, Jonathan Gruber, Joseph P. Newhouse, Heidi Allen, Katherine Baicker, and the Oregon Health Study Group. 2012. "The Oregon Health Insurance Experiment: Evidence from the First Year." *Quarterly Journal of Economics* 127(3):1057–106.
- Goodman-Bacon, Andrew. 2014. "Public Insurance and Mortality: Evidence from Medicaid Implementation." Working paper, University of California, Berkeley.
- Howell, Embry M., and Genevieve M. Kenney. 2012. "The Impact of the Medicaid/CHIP Expansions on Children: A Synthesis of the Evidence." *Medical Care Research and Review* 20 (10):1–25.
- Howell, Embry, Sandra Decker, Sara Hogan, Alshadye Yemane, and Jonay Foster. 2010. "Declining Child Mortality and Continuing Racial Disparities in the Era of the Medicaid and SCHIP Insurance Coverage Expansions." *American Journal of Public Health* 100(12):2500– 506.
- Imbens, Guido W., and Thomas Lemieux. 2008. "Regression Discontinuity Designs: A Guide to Practice." *Journal of Econometrics* 142:615–35.
- Lee, David S., and David Card. 2008. "Regression Discontinuity Inference with Specification Error." *Journal of Econometrics* 142(2):655–74.

- Lee, David S., and Thomas Lemieux. 2010. "Regression Discontinuity Designs in Economics." Journal of Economic Literature 48:281–355.
- Leininger, Lindsey. 2009. "Partial-Year Insurance Coverage and the Health Care Utilization of Children." *Medical Care Research and Review* 66(1):49–67.
- Levy, Helen, and David Meltzer. 2008. "The Impact of Health Insurance on Health." *Annual Review of Public Health* 29:399–409.
- LoSasso, Anthony T., and Thomas C. Buchmueller. 2004. "The Effect of the State Children's Health Insurance Program on Health Insurance Coverage." *Journal of Health Economics* 23 (5):1059–82.
- Manning, Willard G., Joseph P. Newhouse, Naihua Duan, Emmett B. Keeler, and Arleen Leibowitz. 1987. "Health Insurance and the Demand for Medical Care: Evidence from a Randomized Experiment." *The American Economic Review* 77(3):251–77.
- Marton, James, and Aaron Yelowitz. 2014. "Health Insurance Generosity and Conditional Coverage: Evidence from Medicaid and Managed Care in Kentucky." Working paper 2014-5-2, Georgia State University Andrew Young School of Policy Studies.
- Meyer, Bruce D., and Laura R. Wherry. 2012. "Saving Teens: Using a Policy Discontinuity to Estimate the Effects of Medicaid Eligibility." NBER Working Paper 18309.
- Miller, Sarah, and Laura R. Wherry. 2014, "The Long-Term Health Effects of Early Life Medicaid Coverage." Working paper, University of Michigan.
- Racine, Andrew D., Robert Kaestner, Theodore J. Joyce, and Gregory J. Colman. 2001. "Differential Impact of Recent Medicaid Expansions by Race and Ethnicity." *Pediatrics* 108 (5):1135–42.
- Rosenbaum, Sara, D. Richard Mauery, Peter Shin, and Julia Hidalgo. 2005. "National Security and U.S. Child Health Policy: The Origins and Continuing Role of Medicaid and EPSDT." The George Washington University School of Public Health and Health Services, Department of Health Policy, Policy Brief.
- U.S. Census Bureau. 2001. "Table 1. General Mobility, by Region, Sex and Age (Mobility data from March 1999 to March 2000)." Accessed on March 5, 2013. Available at URL http://www.census.gov/hhes/migration/files/cps/p20-538/tab01.txt.
- U.S. House of Representatives. 1996. "1996 Green Book: Background Material and Data on the Programs within the Jurisdiction of the Committee on Ways and Means." Available at URL http://greenbook.waysandmeans.house.gov/.
- Valdez, R. Burciaga, Robert H. Brook, William H. Rogers, John E. Ware, Emmett B. Keeler, Cathy A. Sherbourne, Kathleen N. Lohr, George A. Goldberg, Patricia Camp, and Joseph Newhouse. 1985. "Consequences for Cost-Sharing for Children's Health." *Pediatrics* 75(5):952–61.
- Wherry, Laura R., Sarah Miller, Robert Kaestner, and Bruce D. Meyer. 2015. "Childhood Medicaid Coverage and Later Life Health Care Utilization." NBER Working Paper w20929.